Author's response to reviews

Title: Gene polymorphisms in APOE, NOS3, and LIPC genes may be risk factors for cardiac adverse events after primary CABG

Authors:

Sandra Eifert (seifert@med.uni-muenchen.de)
Astrid Rasch (Astrid.Rasch@med.uni-muenchen.de)
Andres Beiras-Fernandez (abeiras@med.uni-muenchen.de)
Georg Nollert (gnollert@med.uni-muenchen.de)
Bruno Reichart (breichart@med.uni-muenchen.de)
Peter Lohse (plohse@med.uni-muenchen.de)

Version: 6 Date: 17 August 2009

Author's response to reviews:

Dear Sir, please find enclosed the definitive version of the manuscript. I have tried to send it through the official mask, but your system did not allow it.

Yours,

A. Beiras-Fernandez